

This is a repository copy of *Estimating the minimum important difference in the DEMQOL instrument in people with dementia*.

White Rose Research Online URL for this paper: https://eprints.whiterose.ac.uk/175275/

Version: Published Version

Article:

Lee, E.C., Wright, J., Walters, S.J. et al. (2 more authors) (2021) Estimating the minimum important difference in the DEMQOL instrument in people with dementia. Quality of Life Research, 30 (10). pp. 2995-3005. ISSN 0962-9343

https://doi.org/10.1007/s11136-021-02900-7

Reuse

This article is distributed under the terms of the Creative Commons Attribution (CC BY) licence. This licence allows you to distribute, remix, tweak, and build upon the work, even commercially, as long as you credit the authors for the original work. More information and the full terms of the licence here: https://creativecommons.org/licenses/

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.





Estimating the minimum important difference in the DEMQOL instrument in people with dementia

Ellen C. Lee¹ · Jessica Wright¹ · Stephen J. Walters¹ · Cindy L. Cooper¹ · Gail A. Mountain²

Accepted: 27 May 2021 © The Author(s) 2021

Abstract

Purpose The Dementia-Related Quality of Life (DEMQOL) measure and the DEMQOL-Utility Score (DEMQOL-U) are validated tools for measuring quality of life (QOL) in people with dementia. What score changes translate to a clinically significant impact on patients' lives was unknown. This study establishes the minimal important differences (MID) for these two instruments.

Methods Anchor-based and distribution-based methods were used to estimate the MID scores from patients enrolled in a randomised controlled trial. For the anchor-based method, the global QOL (Q29) item from the DEMQOL was chosen as the anchor for DEMQOL and both Q29 and EQ-5D for DEMQOL-U. A one category difference in Q29, and a 0.07 point difference in EQ-5D score, were used to classify improvement and deterioration, and the MID scores were calculated for each category. These results were compared with scores obtained by the distribution-based methods.

Results A total of 490 people with dementia had baseline DEMQOL data, of these 386 had 8-month data, and 344 had 12-month DEMQOL data. The absolute change in DEMQOL for a combined 1-point increase or decrease in the Q29 anchor was 5.2 at 8 months and 6.0 at 12 months. For the DEMQOL-U, the average absolute change at 8 and 12 months was 0.032 and 0.046 for the Q29 anchor and 0.020 and 0.024 for EQ-5D anchor.

Conclusion We present MID scores for the DEMQOL and DEMQOL-U instruments obtained from a large cohort of patients with dementia. An anchored-based estimate of the MID for the DEMQOL is around 5 to 6 points; and 0.02 to 0.05 points for the DEMQOL-U. The results of this study can guide clinicians and researchers in the interpretation of these instruments comparisons between groups or within groups of people with dementia.

Trial Registration Number and date of registration: ISRCTN17993825 on 11th October 2016.

Keywords People with dementia · DEMQOL · DEMQOL · U · Minimum important difference · Responsiveness

Plain english summary

The Dementia-Related Quality of Life questionnaire is one way of measuring the quality of life of people diagnosed with dementia. The questionnaire has 29 questions, about topics including looking after yourself, health and wellbeing and relationships and scores range between 28 and 112. Higher scores indicate better quality of life. Until now, when using this questionnaire for research, we did not know what change or difference in scores a person with dementia would feel was important and which would cause doctors to think about a change in the person's treatment or care. After using statistical techniques to assess this, we estimate that people with dementia would feel that a questionnaire score change of 5 or 6 points would be helpful and would cause their doctors to think about a change in the person's treatment or care.

This information will help researchers to design future trials (they can work out how many people must be recruited) and analyse the results (has the activity or drug had a clinically significant effect on people?).

We also looked at the DEMQOL-Utility score (used by health economists to understand if activities or drugs are value for money) which ranges from 0.243 to 0.986 and is

Ellen C. Lee e.lee@sheffield.ac.uk

¹ Sheffield Clinical Trials Research Unit, School of Health and Related Research, The University of Sheffield, Regent's Court, 30 Regent St, Sheffield S1 4DA, UK

² Centre for Applied Dementia Studies, University of Bradford, Bradford, UK

calculated using the Dementia-Related Quality of Life questionnaire scores. We found that a 0.02 to 0.05 points increase in the DEMQOL-Utility score would indicate that the person being tested was experiencing a clinically significant increase in quality of life.

Introduction

Dementia is an umbrella term for a variety of diseases and conditions affecting the brain distinguished by a decline in memory, language, problem-solving and other thinking skills that affect a person's ability to perform everyday activities. There are a number of sub-types of dementia, but the five most common are Alzheimer's disease, vascular dementia, dementia with Lewy bodies, frontotemporal dementia and mixed dementia. Alzheimer's is the most common cause of dementia [1]. As a person's age increases, so does the risk of them developing dementia [2].

Worldwide, around 50 million people have dementia, and there are up to 10 million new cases every year [3]. Projections show that there will be 82 million people with dementia in 2030 and 152 million in 2050 [3]. Alzheimer's Research UK reports that there are over 850,000 people currently living with dementia in the UK [4]. It is estimated that the number of people with dementia in the UK will rise to over one million by 2021 [1].

There is no treatment currently available to cure dementia or to alter its progressive course. Numerous new treatments are being investigated in various stages of clinical trials [3]. However, much can be offered to support and improve the lives of people with dementia and their careers and families and services are being encouraged to provide post-diagnostic treatment and support [5].

Given that we are currently unable to cure people with dementia, maintaining or enhancing quality of life (QOL) is often an important therapeutic goal [6]. Therefore, accurately measuring QOL is required to both guide clinical decision making and evaluate the impact of various interventions in the population of people with dementia. For this purpose, validated instruments have been developed to evaluate QOL in patients with dementia including the DEMQOL and DEMQOL-U. The DEMQOL is a patient-reported outcome measure (PROM) which is designed to enable the assessment of health-related quality of life of people with dementia [7]. The DEMQOL-U consists of items from the DEMQOL and generates a preference-based single index or score for use in economic evaluations [8].

Interpreting the numerical scores on these instruments can be challenging for a number of reasons. The scales and instruments used may be unfamiliar to many patients and clinicians, who may be uncertain of the meaning of the scale values and summary scores [9]. It is relatively simple to determine the statistical significance of a change in QOL but placing the magnitude of these changes in a context that is meaningful for patients, health professionals, and other stakeholders (pharmaceutical and medical device developers, insurance payers, regulators, governments) has not been so easy. Ascertaining the magnitude of change that corresponds to a minimal important difference would help to address this problem.

Jaeschke et al. 1989 defined the minimal important difference, from the patient perspective, as "The smallest difference in score in the domain of interest which patients perceive as beneficial and which would mandate, in the absence of troublesome side effects and excessive cost, a change in the patient's management" [10].

Thus, individual change standards are needed to provide meaningful interpretation of treatment effects on QOL and to classify patients based on this standard as improved, stable or declined. Anchor-based and distribution-based methods can be used for estimating the MID in QOL instruments [9, 11, 12]. The aim of this study was to estimate the MID for the DEMQOL and DEMQOL-U instruments for comparisons between groups or within groups of people with dementia in a prospective cohort of people with dementia who were recruited as part of an RCT.

Methods

The journeying through dementia trial

The data for this study are provided from the Journeying through Dementia (JtD) Trial, a randomised controlled trial that recruited participants with early-stage dementia from across the UK [13]. In total, 490 participants completed baseline data, 480 of these participants were randomised 1:1 to receive usual care or the JtD programme plus usual care. Participants were eligible for the study if they had a diagnosis of dementia, a Mini Mental State Examination (MMSE) score of 18 or more (indicating that the person was in the milder stages of dementia), a good understanding of English, and were living in the community (not in residential or nursing care). The Journeying through Dementia programme involved twelve weekly group sessions with two trained health practitioners and four individual one-to-one sessions with one of those practitioners. The trial ran 28 groups across 13 sites and those registered for each group varied from four to 12. Participants were followed up at eight and 12 months, and the primary outcome was DEMQOL at eight months post-randomisation.

Instruments and scoring

The DEMQOL questionnaire contains 29 questions [7]. The first 28 items are summed to calculate the DEMOOL score and cover 5 domains: daily activities and looking after yourself, health and well-being, cognitive functioning, social relationships and self-concept. It is completed by the person with dementia, in the case of JtD through interview in the person's home. Each item in the DEMQOL is rated on a 4-point scale: 1 (a lot), 2 (quite a bit), 3 (a little) and 4 (not at all). A separate global quality of life item, Q29, asks respondents to rate 'your quality of life overall' in the past week on 4-point scale, which is rated 1 (very good), 2 (good), 3 (fair) and 4 (poor) (see Online Resource 1 for full wording of Q29). Question 29 does not contribute to the DEMQOL total score. Six items (including Q29) are reverse coded so that for all items, a higher score means better health-related quality of life. Responses to the 28 items on the DEMOOL are summed to generate a total score on a 28 to 112 scale, where higher scores indicate better healthrelated quality of life.

The DEMQOL-U classification system comprises five dimensions (positive emotion, memory, relationships, negative emotion and loneliness) with four levels of increasing impairment associated with each dimension [8]. The scoring algorithms for the DEMQOL-U were derived using the time trade-off elicitation technique in a UK general population sample. The resulting utility scores lie on the zero to one quality-adjusted life years (QALY) scale where zero represents the state dead and one represents the state of full health. The utility scores for the DEMQOL-U range from 0.243 to 0.986.

The EQ-5D-5L (hereinafter referred to as EQ-5D) is a health status classification measured over 5 dimensions: mobility, self-care, usual activities, pain/discomfort and anxiety/depression [14]. Each dimension is assessed by a single question on a five-point ordinal scale (e.g. no problems, slight problems, moderate problems, severe problems, unable). The EQ-5D preference-based measure can be calculated by assigning preference weights value sets to the raw scores. The EQ-5D preference-based score ranges from -0.59 to 1.00 where 1.00 indicates "full health", 0.00 represents dead, and a negative score represents a health status valued as worse than dead.

Statistical analysis

All statistical analyses were performed in Stata v16 [15]. The DEMQOL score was calculated by summing the response to the first 28 items when at least half (14 or more) of the items were answered, the missing items were imputed with the mean of the completed items [16]. The EQ-5D was scored using the mapping function developed by van Hout

et al. [17], no score was calculated if any items were missing. Summaries and analysis were presented on JtD participants that had at least baseline data (n = 490), this is more than those randomised as a few participants withdrew prior to randomisation.

Both anchor-based and distribution-based methods were used to determine the MID as recommended by Revicki et al. [12]. For the anchor-based methods, the response to question 29 of the DEMQOL (DEMQOL Q29), which asks "... in the last week, how would you rate your quality of life overall" was considered as an anchor for DEMQOL as it is a self-reported single question rating of overall quality of life. The four responses to this item are very good, Good, Fair and Poor. The DEMQOL Q29 and EQ-5D were considered as anchors for DEMQOL-U. Spearman's correlation coefficient was calculated to assess the correlation between the measures and the anchors with values over 0.3 deemed as having acceptable association between anchor and the outcome measure [12]. For the cross-sectional anchor analysis, baseline DEMQOL and DEMQOL-U were summarised over the four DEMQOL Q29 response categories (Quality of life overall: very good, good, fair and poor). One-way ANOVA was used to test for difference in baseline DEMQOL and DEMQOL-U means across the four Q29 response groups.

For the longitudinal anchor, change scores were calculated for each outcome for baseline to 8 months and baseline to 12 months. A change of 0.07 in EQ-5D score was used to classify a negative or a positive change; this was guided by an observed mean MID of 0.074 over 11 studies from a review by Walters and Brazier [9]. Participants with a less than 0.07 absolute change, in the EQ-5D score, were classified as no change. For the DEMQOL, change over time from baseline to follow-up was summarised over the seven possible Q29 change categories. The DEMQOL-U change was summarised over both Q29 and EQ-5D response categories. The negative and positive change scores were combined to provide a single MID estimate when the MID evidence was consolidated (for example, a 1 category improvement with the absolute values of a 1 category deterioration in Q29 scores) hence assuming the cohorts being combined are identical except for the sign.

All distribution-based methods are sample dependent (as they use sample SD) and are all considered as proxies to MID as they do not provide information about a difference that is considered minimally important [18]. Hence, they were used to provide supporting evidence of the MID and compared with the anchor-based MID scores. The previously observed effect size was reported, alongside SD per treatment group at baseline for studies that used DEMQOL. Studies using DEMQOL were identified through literature searches on Medline and EMBASE for randomised controlled trials reported in English in any population type. Studies solely using DEMQOL-Proxy, or a translated version of DEMQOL, were not included. The search also highlighted several observational studies using DEMQOL that were included in the results either due to their large scale or relevance.

The standardised effect size was calculated using the between-person SD at baseline from the JtD data for small (0.2 SD) to medium (0.5 SD) effect sizes [19]. Smith et al. [7] found the test-retest reliability (intra-class correlation) of DEMQOL to be 0.84 (n=17) amongst the whole sample and 0.76 (n=10) amongst the subset with mild/moderate MCI MMSE \geq 10. Reliability (Internal consistency—Cronbach's alpha) was found to be 0.87 (n=75) amongst the whole sample and 0.87 (n=68) amongst those with MMSE \geq 10. The standard error of measurement (SEM [20]) was calculated using the between-person SD at baseline and scale reliability (r) observed by Smith et al. [7] (SEM = SD $\sqrt{(1-r)}$). The SEM was calculated using both test-retest reliability and Cronbach's α .

Results

A total of 490 participants were recruited to the JtD study and had available baseline DEMQOL data, of these 386 also had 8-month data, and 344 had both baseline and 12-month DEMQOL data. Table 1 summarises the characteristics of the cohort. The median age of the cohort was 78.0 and 58% of the participants were male. The most prevalent diagnosis was Alzheimer's (60%), the median length of time since dementia was diagnosed was 0.7 years (IQR 0.3-1.8 years). The inclusion criteria for this study were a MMSE score of 18 or more, so all participants in the cohort had either mild cognitive impairment (38%) or normal cognitive function (62%) at baseline according to the MMSE. The characteristics are presented separately for those that had baseline data, baseline and 8-month data, and baseline and 12-month data, the three groups appear to have similar characteristics, with a slightly lower proportion of participants with mild cognitive impairment in the group with 12-month data available. Table 2 shows how the mean DEMQOL and DEMQOL-U scores changed over time for the whole cohort and the subsample who completed all three assessments. Overall, there was little change in DEMQOL and DEMQOL-U scores over time.

All correlations between DEMQOL(-U) and the anchors at each timepoint are above the threshold of 0.3 and so the anchors have an appreciable association with the DEMQOL and DEMQOL-U (see Online Resource 2). DEMQOL-U has slightly higher correlation with Q29 than it does with EQ-5D. The correlation between change scores at 8 and 12 months for DEMQOL-U with both EQ-5D and Q29 was below the threshold of 0.3; EQ-5D was particularly low so the longitudinal anchor analysis for DEMQOL-U and EQ-5D is to be treated cautiously.

Anchor-based methods

The distributions of DEMQOL and DEMQOL-U at baseline over the four Q29 quality of life overall categories are displayed in Table 3. The mean DEMQOL score is eight points higher in the participants that chose good health compared to fair, and the difference is similar in those that chose very good compared to good. The difference in DEMQOL-U between categories was around 0.06 (0.064 between fair and good, 0.057 between good and very good). The oneway ANOVA for the difference in means across the four categories was statistically significant (p < 0.001) for both DEMQOL and DEMQOL-U.

The longitudinal anchor analysis is presented in Table 4. A one category increase in DEMQOL Q29 corresponds with a 7.2 average increase at 8 months and a 7.7 increase at 12 months in DEMQOL score. The absolute difference in DEMQOL is smaller for a one category decrease in Q29 at both 8 months and 12 months (-3.2 change at 8 months and)-4.3 change at 12 months). Combining the two categories results in an average 5.2 point absolute change at 8 months and 6.0 point change at 12 months. The absolute change in DEMQOL-U for a 1-point increase or 1-point decrease in Q29 ranges from 0.023 to 0.054, and DEMQOL-U change related to a positive or negative change in EQ-5D ranges from 0.007 to 0.027 in absolute value. The absolute values of the mild improvement and deterioration groups were combined; these resulted in an average absolute change at 8 and 12 months being 0.032 and 0.046 for the Q29 anchor and 0.020 and 0.024 for EQ-5D anchor.

Distribution based methods

Previously observed effect size

The JtD trial compared DEMQOL outcome between the treatment groups at 8 or 12 months post-randomisation. In a review of other randomised controlled trials that used DEMQOL, we found 10 RCTs using DEMQOL as an outcome: eight of these had published results [21–28] and two were ongoing [29, 30]. Of the eight RCTs with published results, one used DEMQOL as the primary outcome [27]. Table 5 shows that one trial observed a statistically significant treatment difference of 7.4; however, the sample size was small and the standard deviations observed were not in line with the other studies [27]. Most of the observed standard deviations at baseline are between 12 and 15 which are in keeping with our observed SD of 13.1 across treatment groups.

Characteristic	With baseline data $(n = 490)$	With 8 m data ($n = 386$)	With 12 m data $(n=344)$	
Sex				
Male	283 (57.8%)	227 (58.8%)	206 (59.9%)	
Female	207 (42.2%)	159 (41.2%)	138 (40.1%)	
Age				
n (%)	490 (100.0%)	386 (100.0%)	344 (100.0%)	
Mean (SD)	77.1 (7.4)	76.4 (7.5)	76.4 (7.4)	
Median (IQR)	78.0 (73.0, 83.0)	77.0 (72.0, 82.0)	77.0 (72.0, 81.0)	
Min., Max	39.0, 93.0	39.0, 93.0	39.0, 90.0	
Ethnicity				
English/Welsh/Scottish/Northern Irish/British	468 (95.5%)	368 (95.3%)	325 (94.5%)	
Irish	7 (1.4%)	5 (1.3%)	6 (1.7%)	
Any other White background	5 (1.0%)	4 (1.0%)	4 (1.2%)	
Indian	3 (0.6%)	3 (0.8%)	3 (0.9%)	
Other	6 (1.2%)	6 (1.6%)	6 (1.7%)	
Unknown	1 (0.2%)	0 (0.0%)	0 (0.0%)	
Lives with others				
No	129 (26.3%)	92 (23.8%)	84 (24.4%)	
Yes	360 (73.5%)	293 (75.9%)	259 (75.3%)	
Type of dementia diagnosed				
Alzheimer's	296 (60.4%)	242 (62.7%)	216 (62.8%)	
Vascular dementia	50 (10.2%)	34 (8.8%)	29 (8.4%)	
Mixed Alzheimer's/vascular dementia	112 (22.9%)	85 (22.0%)	77 (22.4%)	
Dementia in Parkinson disease	6 (1.2%)	5 (1.3%)	4 (1.2%)	
Frontotemporal dementia (FTD)	7 (1.4%)	6 (1.6%)	5 (1.5%)	
Lewy body dementia	4 (0.8%)	3 (0.8%)	3 (0.9%)	
Unspecified dementia	12 (2.4%)	10 (2.6%)	9 (2.6%)	
Other	2 (0.4%)	1 (0.3%)	1 (0.3%)	
Length of time since dementia diagnosis (years)				
n (%)	479 (97.8%)	385 (99.7%)	343 (99.7%)	
Mean (SD)	1.3 (1.6)	1.3 (1.6)	1.4 (1.7)	
Median (IQR)	0.7 (0.3, 1.8)	0.8 (0.3, 1.8)	0.8 (0.3, 1.9)	
Min., Max	0.0, 13.0	0.0, 13.0	0.0, 13.0	
MMSE cognitive impairment				
Mild	184 (37.6%)	128 (33.2%)	113 (32.8%)	
Normal	306 (62.4%)	258 (66.8%)	231 (67.2%)	
MMSE (total score)				
n (%)	490 (100.0%)	386 (100.0%)	344 (100.0%)	
Mean (SD)	24.6 (3.1)	24.9 (3.1)	25.0 (3.1)	
Median (IQR)	25.0 (22.0, 27.0)	25.0 (23.0, 27.0)	25.0 (23.0, 28.0)	
Min., Max	18.0, 30.0	18.0, 30.0	18.0, 30.0	

Standardised effect size

Standard error of measurement

Cohen [19] defined a 0.2 standardised effect size as small and 0.5 as medium. Using the between-person SD at baseline, of 13 points, a small standardised effect size in DEMQOL is 2.6 points and a medium 6.5 points (Table 6). Using the estimates of reliability for participants with $MMSE \ge 10$, the SEMs in Table 7 are calculated; 6.37 for test-retest reliability and 4.69 for internal consistency.

Table 2DEMQOL andDEMQOL-U summaries bytimepoint for JtD participants(N = 490)

	DEMQO	DEMQOL			DEMQOL-U		
	n	Mean	SD	n	mean	SD	
Participants with	available data						
Baseline	490	90.7	13.0	490	0.862	0.114	
8 months	388	92.6	13.8	386	0.87	0.120	
12 months	352	92.0	14.1	345	0.872	0.117	
Participants with	data at all thre	e timepoints					
Baseline	337	90.9	13.4	337	0.865	0.118	
8 months	337	92.2	14.1	337	0.868	0.124	
12 months	337	92.1	14.0	337	0.872	0.117	

 Table 3
 DEMQOL and DEMQOL-U summaries at baseline, across

 DEMQOL Q29 categories (cross-sectional anchor)

Baseline	n	Mean	SD	95% CI for mean
DEMQOL				
Q29 Response				
Poor	10	61.6	16.1	50.5-73.2
Fair	89	81.8	11.9	79.3-84.3
Good	239	89.8	11.3	88.3-91.2
Very good	152	99.2	8.3	97.9-100.5
DEMQOL-U				
Q29 Response				
Poor	10	0.544	0.197	0.404-0.685
Fair	89	0.799	0.117	0.774-0.823
Good	239	0.863	0.092	0.851-0.875
Very good	152	0.920	0.080	0.907-0.933

ANOVA *p* < 0.001

Combining estimates

Figure 1 shows the distribution and anchor-based estimates of MID for both DEMQOL and DEMQOL-U. The longitudinal anchor-based methods for improvement and deterioration are presented alongside the absolute change pooled for both improvement and deterioration groups. A MID of 5 is consistent with the longitudinal anchor MID estimate and is centred amongst the distribution-based methods, it corresponds to a 0.38 standardised effect size. EQ-5D anchor-based methods for DEMQOL-U have been excluded from the summary (Fig. 1); due to the low correlation between change EQ-5D and DEMQOL-U change scores, the estimates will be unreliable and likely to underestimate MID. The DEMQOL Q29-based anchor methods for DEMQOL-U fall in line with 0.3–0.5 SD estimate range.

Discussion

We have calculated distribution, anchor, and standard error of measurement-based estimates of the MID for the DEMQOL and DEMQOL-U instruments using a large cohort of 490 patients with dementia. We found that the anchored-based estimate of the MID for the DEMQOL is around 5–6 points; and it is 0.02–0.05 points for the DEMQOL-U for comparisons between groups or within groups of people with dementia.

The minimum detectable change (MDC) has been defined as the smallest change in score (at an individual level) that can be detected after allowing for measurement error of the instrument. There are several methods for estimating the MDC, usually involving the standard error of measurement (SEM) calculated from reliability coefficients such as test–retest and Chronbach's alpha [11]. We used two methods two estimate the SEM (test–retest reliability and internal consistency) and calculated values of 6.4 and 4.7 for one SEM, respectively, from these methods.

Ideally, for a sensitive and reliable instrument, the MDC should be smaller than the MID [11]. The MID and MDC are important but they are different concepts measuring different things. The relationship between the MID and MDC is discussed by Turner et al. 2010 and de Vet (2010) [33, 34]. They agree that the MDC (and the associated SEM) cannot reliability replace the MID. They conclude that the MDC is a statistical property of the measurement, but the MID is the value of concern for interpretation and is based on the judgement of patients. The MDC (and the associated SEM) is of little relevance for interpretation [11].

We observed that the MID value from the anchor-based assessment of 5 was smaller than the SEM of 6.4 (based on test-retest reliability). In these circumstances, changes as large as the MID may be important for patients, but they cannot be distinguished from measurement error [34]. We also observed that the MID of 5 was slightly larger the then SEM of 4.7 (based on internal consistency reliability). In this situation, changes as large as the MID can be considered statistically significant and important to patients [34].

	n	Mean	SD	95% CI for mean	
DEMQOL chang	e Baseli	ne to 8 mont	ths		
Q29 change					
- 3	3	- 22.8	18.6	- 69.1-23.5	
- 2	6	- 4.2	10.3	- 15.0-6.7	
- 1	80	- 3.2	9.6	- 5.3 to - 1.0	
0	210	1.6	9.1	0.3-2.8	
1	81	7.2	9.8	5.0-9.4	
2	7	11.4	10.6	1.6-21.3	
3	1	35.0			
DEMQOL chang	e Baseli	ne to 12 mor	nths		
Q29 change					
- 2	12	- 8.8	10.3	- 15.3 2.3	
- 1	66	- 4.3	10.1	- 6.7 to - 1.8	
0	199	1.3	9.0	0.0–2.5	
1	69	7.7	9.6	5.4-10.0	
2	3	12.7	4.0	2.6-22.7	
DEMQOL- U cl	hange B	aseline to 8 r	nonths		
Q29 change					
- 3	2	- 0.063	0.262	- 2.420-2.295	
- 2	6	- 0.075	0.066	-0.145 to -0.006	
- 1	80	- 0.023	0.109	-0.047 - 0.001	
0	209	0.002	0.114	- 0.013-0.018	
1	81	0.041	0.090	0.021-0.061	
2	7	0.078	0.071	0.012-0.144	
3	1	0.545			
EQ-5D change					
Negative*	94	-0.007	0.106	- 0.028-0.015	
No change*	165	0.007	0.101	- 0.008-0.023	
Positive*	124	0.017	0.131	- 0.006-0.040	
DEMQOL-U change Baseline to 12 months					
Q29 change					
- 2	12	- 0.045	0.085	- 0.099-0.009	
- 1	65	- 0.039	0.096	-0.063 to -0.015	
0	196	- 0.005	0.093	-0.008-0.018	
1	68	0.054	0.114	0.026-0.081	
2	3	0.139	0.102	- 0.033-0.311	
EQ-5D change					
Negative*	87	- 0.017	0.104	-0.040 - 0.005	
No change*	140	0.006	0.103	- 0.011-0.023	
Positive*	113	0.027	0.096	0.009-0.045	

Table 4DEMQOL and DEMQOL-U change by DEMQOL Q29 andEQ-5D change categories (longitudinal anchor)

*Negative change defined as ≤ 0.07 , no change defined as between -0.07 and 0.07, positive change defined as > 0.07

To our knowledge, this study is the first to elicit and make recommendations on the MID for DEMQOL and DEMQOL-U. Both anchor and distribution-based methods were used, and results were consolidated. Anchor-based analysis was only conducted for anchors that achieved moderate correlation with the measure of choice ($|r| \ge 0.3$). Using EQ-5D as a longitudinal anchor may have overestimated the MID estimate of DEMQOL-U as it included participants with a large absolute change in the estimate however, the estimates that were observed were smaller than using DEMQOL Q29, this was likely due to the low correlation between change scores, producing unreliable, and underestimated MID estimates.

Using global OOL as an anchor is one accepted method and has been used in previous studies [9, 35, 36]. The advantages of using a global rating of QOL as an anchor are that they are relatively easy to obtain, patient-centred and can take into account a variety of information and determinants of well-being. However, in order to use an anchor-based method, there must exist some association (minimum correlation) between the QOL items and the chosen anchor [12]. A potential limitation of our study is that some of the correlations between the anchors used in our study, and the DEMOOL and DEMOOL-U scores were not that strong (the correlations ranged from 0.38 to 0.59), with the possible result that some of the anchors may not perform well in defining the MID. Nonetheless, these correlations were above the recommended minimum thresholds of 0.30 and the even higher value of 0.371 recommended by Hays et al. [37].

Other potential limitations associated with using DEMQOL Q29 as an anchor include recall bias and response shift [11]. Response shift occurs when a patient's views, values or expectations change over time. Thus, a patient's health might be seen to be getting worse, yet the patient with dementia may assert that their QOL has not changed, or even that it has improved. Alternatively, a patient's health status may appear to be unchanging even though that same patient may report substantial changes in their QOL. It is difficult to assess response shift in people with dementia. The classical way of assessing response shift is to use the "then test" which may not be appropriate for patients with dementia. The then-test usually asked patients after completing a second assessment a question such as "We would like you to think back to the time of your first assessment. With hindsight, how would you now rate the way you felt then?" [11]. The difference between the values of the 1st assessment and the then-test provides an estimate of response shift.

Recall bias occurs when the respondent's answer to a question is affected by the respondent's memory. People tend to forget how extreme the past was. Response is likely to be influenced by the patient's status at the time of recall. For these reasons, items with short recall periods or items that ask patients to describe their current or recent status are preferable [11]. The DEMQOL asks respondents questions about how they feel "right now" or "in the last week" so has a short recall period. However, response shift and recall bias remain a possibility with people with dementia, and **Table 5**Standard deviation andobserved treatment differencefor studies using DEMQOL

Study	Observed n with outcome data (total) ^a	SD per group at baseline	Observed absolute treatment differ- ences
Trials			
Journeying through Dementia	381	13.0, 13.2	Not yet published
HTA SADD trial ^b [21]	456, 405	14.3, 12.8, 17.2	0.03-1.76
The CSP-RYCT trial [22]	138	12.4, 13.4, 11.2	0.173-2.54
SMILE study [23]	371, 343	15.5, 13.8	0.1, 0.05
Individual cognitive stimulation therapy for dementia [24]	227, 264	11.76, 13.55	0.33, 1.73
Maintenance cognitive stimula- tion therapy [25]	218, 199	10.9, 11.7	0.3, 0.86
The GREAT trial [26]	445, 417	12.9, 11.75	0.24, 1.08
Crotty et al. hip fracture [27]	90, 70	27.8, 8.48	0.3–7.4
NIHR TEAM study [28]	222	11.9, 13.4	0.7
Observational studies			
Park et al. 2017 [31]	785	12.5	n/a
Banerjee 2006 ^c [32]	101	13.75	n/a

^anumbers separated by a comma are n at first follow-up and n at second follow-up

^bDEMQOL data completion unknown, *n* is any outcome data

^cn in cohort

 Table 6
 Standardised effect sizes using a baseline SD of 13 points

	DEMQOL		DI	EMQOL-U		
0.2 SD	2.6		0.0	023		
0.3 SD	3.9	3.9		0.034		
0.4 SD	5.2	5.2		0.046		
0.5 SD	6.5	6.5		0.057		
Table 7 SEM c	alculations	SD	r	SEM: SD × $\sqrt{(1-r)}$		
		13	0.76 (Test– retest reli- ability)	6.37		
		13	0.87 (Inter- nal consist- ency)	4.69		

this is a limitation. Indeed, given our patient population, our study is more likely to suffer from these possible threats to the validity of the results than other studies using similar anchor-based approaches.

We used a four-point global QOL anchor which resulted in seven-point change scale; others have used 14 points, which may be more sensitive [10]. The designation of what response on the DEMQOL Q29 anchor which suggests patients as fundamentally unchanged and what response suggests patients have experienced a small but important change is inevitably subjective.

The anchor, DEMQOL Q29, did not ask the participant to recall their overall health but instead rate it considering the previous week; this is a strength of the anchor and the global rating of QoL. However, a limitation of the study is that people with dementia had to self-report their quality of life and be able to do so again 12 months later, in which time their dementia may have progressed to moderate stages, and therefore, self-report could be less reliable for some participants. Only 70% (344/490) of participants completed the 12-month assessment; some of these may have been due to declining health.

This study was conducted in participants with mild/moderate dementia and in the relatively early stages of the disease, living in the community in the United Kingdom. Many (62%) of the participants were assessed as having normal cognitive function on the MMSE at baseline. The recommended MID estimates are most applicable to studies in a similar patient population-a single MID is unlikely to be relevant for all populations, and a different MID may be appropriate for later-stage dementia and participants living in residential or nursing care. A limitation of the study is that 21% of the participants were excluded from the longitudinal anchor analysis due to having missing outcome data (either having withdrawn from the study, lost to follow-up or death); however, Table 1 shows baseline characteristics to be similar for the subset that had available eight- and twelve-month data compared to all those that had baseline data.

Fig. 1 Summary of anchor and distribution-based Minimally Important Difference (MID) estimates for DEMQOL and DEMQOL-U method



Second, the follow-up duration was only 12 months, so the MID values that were determined herein may not be consistent with those in studies with longer follow-up durations. Thus, to validate the findings of this study, a longer follow-up and patients with different severities of dementia would be needed in future studies.

Our recommended MID should be confirmed based on evidence from other studies, and we recommend that other cohorts are used to estimate MID and the evidence is consolidated to provide a more robust estimate. Further research into the MID for the DEMQOL-proxy is needed for people with dementia who cannot self-report.

Conclusion

In this study, we established MID scores for DEMQOL and DEMQOL-U using both anchor and distributionbased methods on a prospective cohort of people living with dementia. We found that the anchor-based estimate of the MID for the DEMQOL is around 5–6 points; and 0.02–0.05 for the DEMQOL-U. This study can guide researchers when designing studies and calculating sample sizes and when interpreting DEMQOL(-U) change over time and treatment differences for comparisons between groups or within groups of people with dementia.

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s11136-021-02900-7. Acknowledgements We wish to acknowledge all Journeying through Dementia trial staff, participants and members of the trial review boards—the Data Monitoring and Ethics Committee and Trial Steering Committee. In particular, we wish to thank the Trial Chief Investigator Professor Gail Mountain, University of Bradford and the Director of Sheffield Clinical Trials Unit, Professor Cindy Cooper for their support.

Author contributions The study was designed by EL and SW Literature reviews underpinning the work were conducted by EL, SW and JW. Data analysis was conducted by EL The first draft of the manuscript was written by EL with contributions from SW, JW, CC and GM. All authors read and approved the final manuscript.

Funding This study was funded under the grant awarded to the Journeying through Dementia clinical trial by the National Institute for Health Research (NIHR) Health Technology Assessment (HTA) Programme grant number 14/140/80. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health. Further information available at: https://fundingawards.nihr.ac.uk/award/14/140/80.

Data availability All data requests should be submitted to the corresponding author for consideration. Access to anonymised data may be granted following review.

Code availability All code requests should be submitted to the corresponding author for consideration.

Declarations

Conflicts of interest SW reports grants from other NIHR HTA Programme during the conduct of the study; grants from other research funders, personal fees from Book Royalties and personal fees from external examining, outside the submitted work. EL and JW declare that they have no conflict of interest.

Ethical approval This study was performed as part of the wider Journeying through Dementia (JtD) clinical trial. The JtD trial was approved by Leeds East Research Ethics Committee, on 01/07/16, reference number 16/YH/0238. The trial was conducted in line with the principles of the Declaration of Helsinki. Informed consent was requested and received from all study participants and recorded on documents approved by the Research Ethics Committee.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- 1. Alzheimer's Association (2020). What Is Dementia? Alzheimer's Association. https://www.alz.org/alzheimers-dementia/what-is-dementia. Accessed 28 May 2020
- Prince, M., Knapp, M., Guerchet, M., McCrone, P., Prina, M., Comas-Herrera, A., et al. (2014). *Dementia UK Update* (2nd ed.). London: Alzheimer's Society.
- World Health Organization (2020). *Dementia*. World Health Organization. https://www.who.int/news-room/fact-sheets/detail/ dementia. Accessed 28 May 2020
- Alzheimer's Research UK (2018). Numbers of people in the UK. https://www.dementiastatistics.org/statistics/numbers-of-peoplein-the-uk/. Accessed 7 Oct 2019
- NHS England (2016). Well pathway for dementia. https://www. england.nhs.uk/mentalhealth/wp-content/uploads/sites/29/2016/ 03/dementia-well-pathway.pdf. Accessed 30 Oct 2018
- Winblad, B., Amouyel, P., Andrieu, S., Ballard, C., Brayne, C., Brodaty, H., et al. (2016). Defeating Alzheimer's disease and other dementias: A priority for European science and society. *Lancet Neurology*, 15, 455–532.
- Smith, S., Lamping, D., Banerjee, S., Harwood, R., Foley, B., Smith, P., et al. (2005). Measurement of health-related quality of life for people with dementia: Development of a new instrument (DEMQOL) and an evaluation of current methodology. *Health Technology Assessment*, 9(10), 1–93.
- Mulhern, B., Rowen, D., Brazier, J., Smith, S., Romeo, R., Tait, R., et al. (2013). Development of DEMQOL-U and DEMQOL-PROXY-U: Generation of preference-based indices from DEMQOL and DEMQOL-PROXY for use in economic evaluation. *Health Technology Assessment*, 17(5), 1–140.
- 9. Walters, S. J., & Brazier, J. E. (2005). Comparison of the minimally important difference for two health state utility measures: EQ-5D and SF-6D. *Quality of Life Research*, 14(6), 1523–1532.
- Jaeschke, R., Singer, J., & Guyatt, G. H. (1989). Measurement of health status. Ascertaining the minimal clinically important difference. *Controlled Clinical Trials*, 10(4), 407–415.

- 11. Fayers, P. M., & Machin, D. (2016). *Quality of life : The assessment, analysis, and reporting of patient-reported outcomes* (3rd ed.). Chichester: Wiley-Blackwell.
- Revicki, D., Hays, R. D., Cella, D., & Sloan, J. (2008). Recommended methods for determining responsiveness and minimally important differences for patient-reported outcomes. *Journal of Clinical Epidemiology*, *61*, 102–109.
- Wright, J., Foster, A., Cooper, C., Sprange, K., Walters, S., Berry, K., et al. (2019). Study protocol for a randomised controlled trial assessing the clinical and cost-effectiveness of the Journeying through Dementia (JtD) intervention compared to usual care. *British Medical Journal Open*, 9(9), e029207.
- 14. EuroQol Research Foundation (2019). EuroQol research foundation. EQ-5D-5L User Guide
- 15. StataCorp (2019). Stata statistical software: Release 16
- Banerjee S. (2019). DEMQOL: Dementia quality of life measure. In: Bright Sussex Med Sch. https://www.bsms.ac.uk/research/ neuroscience/cds/research/demqol.aspx. Accessed 15 Oct 2019
- van Hout, B., Janssen, M. F., Feng, Y.-S., Kohlmann, T., Busschbach, J., Golicki, D., et al. (2012). Interim scoring for the EQ-5D-5L: Mapping the EQ-5D-5L to EQ-5D-3L value sets. Value in health the journal of the International Society for Pharmaco-economics and Outcomes Research, 15(5), 708–715.
- King, M. T. (2011). A point of minimal important difference (MID): A critique of terminology and methods. *Expert Review of Pharmacoeconomics and Outcomes Research*, 11(2), 171–184.
- 19. Cohen, J. (1988). *Statistical power analysis for the behavioral sciences* (2nd ed.). Hillsdale, NJ: Lawrence Earlbaum Associates.
- Jacobson, N. S., & Truax, P. (1991). Clinical significance: A statistical approach to defining meaningful change in psychotherapy research. *Journal of Consulting and Clinical Psychology*, 59(1), 12–19.
- Banerjee, S., Hellier, J., Romeo, R., Dewey, M., Knapp, M., Ballard, C., et al. (2013). Study of the use of antidepressants for depression in dementia: The HTA-SADD trial- A multicentre, randomised, double-blind, placebo-controlled trial of the clinical effectiveness and cost-effectiveness of sertraline and mirtazapine. *Health Technology Assessment*, 17(7), 1–43.
- Charlesworth, G., Burnell, K., Crellin, N., Hoare, Z., Hoe, J., Knapp, M., et al. (2016). Peer support and reminiscence therapy for people with dementia and their family carers: A factorial pragmatic randomised trial. *Journal of Neurology Neurosurgery and Psychiatry*, 87(11), 1218–1228.
- Low, L.-F., Brodaty, H., Goodenough, B., Spitzer, P., Bell, J.-P., Fleming, R., et al. (2013). The sydney multisite intervention of laughter bosses and elder clowns (SMILE) study: Cluster randomised trial of humour therapy in nursing homes. *British Medical Journal Open*, 3(1), 1–8.
- Orgeta, V., Leung, P., Yates, L., Kang, S., Hoare, Z., Henderson, C., et al. (2015). Individual cognitive stimulation therapy for dementia: A clinical effectiveness and cost-effectiveness pragmatic, multicentre, randomised controlled trial. *Health Technology Assessment*, 19(64), 7–73.
- Orrell, M., Aguirre, E., Spector, A., Hoare, Z., Woods, R. T., Streater, A., et al. (2014). Maintenance cognitive stimulation therapy for dementia: Single-blind, multicentre, pragmatic randomised controlled trial. *British Journal of Psychiatry*, 204(6), 454–461.
- Clare, L., Kudlicka, A., Oyebode, J. R., Jones, R. W., Bayer, A., Leroi, I., et al. (2019). Goal-oriented cognitive rehabilitation for early-stage Alzheimer's and related dementias: The GREAT RCT. *Health technology assessment (Winchester, England, 23*(10), 1–242.
- 27. Crotty, M., Killington, M., Liu, E., Cameron, I. D., Kurrle, S., Kaambwa, B., et al. (2019). Should we provide outreach rehabilitation to very old people living in nursing care facilities after a hip

fracture? A randomised controlled trial. *Age and ageing*, 48(3), 373–380.

- Goldberg, S. E., Bradshaw, L. E., Kearney, F. C., Russell, C., Whittamore, K. H., Foster, P. E. R., et al. (2013). Care in specialist medical and mental health unit compared with standard care for older people with cognitive impairment admitted to general hospital: Randomised controlled trial (NIHR TEAM trial). *BMJ*, 347, 7917.
- Wenborn, J., Hynes, S., Moniz-Cook, E., Mountain, G., Poland, F., King, M., et al. (2016). Community occupational therapy for people with dementia and family carers (COTiD-UK) versus treatment as usual (Valuing Active Life in Dementia [VALID] programme): Study protocol for a randomised controlled trial. *Trials*, 17(1), 1–10.
- Goldberg S. (2020). ISRCTN15320670: Promoting activity, independence and stability in early dementia and mild cognitive impairment 2. In: ISRCTN. http://www.isrctn.com/ISRCTN1532 0670. Accessed 14 July 2020
- Park, M. H., Black, N., Ritchie, C. W., Hendriks, A. A. J., & Smith, S. C. (2017). Is the effectiveness of memory assessment services associated with their structural and process characteristics? *International Journal of Geriatric Psychiatry*, 33(1), 75–84.
- Banerjee, S., Smith, S. C., Lamping, D. L., Harwood, R. H., Foley, B., Smith, P., et al. (2006). Quality of life in dementia: more than just cognition. An analysis of associations with quality of life in dementia. *Journal of Neurological Neurosurgery and Psychiatry*, 77(2), 146–148.

- Turner, D., Schünemann, H. J., Griffith, L. E., Beaton, D. E., Griffiths, A. M., Critch, J. N., et al. (2010). The minimal detectable change cannot reliably replace the minimal important difference. *Journal of Clinical Epidemiology*, 63(1), 28–36.
- de Vet, H. C. W., & Terwee, C. B. (2010). The minimal detectable change should not replace the minimal important difference. *Journal of Clinical Epidemiology*, 63(7), 804–805.
- Alanne, S., Roine, R. P., Räsänen, P., Vainiola, T., & Sintonen, H. (2015). Estimating the minimum important change in the 15D scores. *Quality of Life Research*, 24(3), 599–606.
- 36. Raman, S., Ding, K., Chow, E., Meyer, R. M., van der Linden, Y. M., Roos, D., et al. (2018). Minimal clinically important differences in the EORTC QLQ-C30 and brief pain inventory in patients undergoing re-irradiation for painful bone metastases. *Quality of Life Research*, 27(4), 1089–1098.
- Hays, R. D., Farivar, S. S., & Liu, H. (2005). Approaches and recommendations for estimating minimally important differences for health-related quality of life measures. *COPD Journal of Chronic Obstructive Pulmonary Disease*, 2(1), 63–67.

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.